Author's response to reviews

Title: Consultant medical trainers, modernising medical careers (MMC) and the European time directive (EWTD): tensions and challenges in a changing medical education context.

Authors:

Maria Tsouroufli (M.Tsouroufli@uea.ac.uk)
Heather Payne (payneeh@cf.ac.uk)

Version: 5 Date: 9 May 2008

Author's response to reviews:

Dear editors,
The list of authors in the manuscript are now exactly as they are in the submission system.
Affiliations are written in full and each author is linked to their corresponding affiliation with the use of superscript numbers.
Email addresses are separated from affiliations and are on a new line.
I have also removed websites from the text.
'Abbreviations' is provided in a paragraph.
Authors' contributors has been changed to authors contributions.